Author's response to reviews

Title: The Acceptability and Feasibility of an Intercultural Birth Center in the Highlands of Chiapas, Mexico

Authors:

   Kathryn M Tucker (katy.m.tucker@gmail.com)
   Hector Ochoa (hochoa@ecosur.mx)
   Rosario Garcia (rgmiranda7@gmail.com)
   Kirsty Sievwright (kms289@georgetown.edu)
   Amy Chambliss (ama243@georgetown.edu)
   Margaret C Baker (mcb93@georgetown.edu)

Version: 6  Date: 21 March 2013

Author's response to reviews: see over
Dear Dr Crow,

Please find attached the revised article with the following paragraph added:

This study was conducted under the auspices of the El Colegio de la Frontera Sur and according to Mexican regulations it was considered as exempt from IRB review due to the non invasive methods used. Consent was obtained verbally from all participants.

Best wishes,

Dr Margaret Baker